Author’s response to reviews

Title: A Rare Case of Pericarditis and Pleural Empyema Secondary to Transdiaphragmatic Extension of Pyogenic Liver abscess

Authors:

Eunae Cho (cea202@hanmail.net)
Sang Woo Park (withpsw@naver.com)
Chung Hwan Jun (estevanj@naver.com)
Sang Soo Shin (kjradsse@jnu.ac.kr)
Eun Kyu Park (iameunkyu@gmail.com)
Kyo Seon Lee (waytogosun@naver.com)
Seon Young Park (drpsy@naver.com)
Chang Hwan Park (p1052ccy@hanmail.net)
Hyun Soo Kim (dshskim@jnu.ac.kr)
Sung Kyu Choi (choisk@jnu.ac.kr)
Jong Sun Rew (jsrew@jnu.ac.kr)

Version: 2 Date: 28 Dec 2017

Author’s response to reviews:

Point-by-point response letter

Did the patient present with any ophthalmological symptoms and was the patient referred to the ophthalmologists for examination of the eyes? Please write a short paragraph on the associated ophthalmological complications.

Thank you for your comments. The patient did not have any ophthalmological symptoms (we added this in page 4, line 16-17), and was not referred to the ophthalmologists for eye examination. He has not complained of decreased visual acuity after discharge until now. And we have inserted a paragraph about ophthalmological complication in page 8, line 22~ page 9, line 5.